Author's response to reviews

Title: Improvements to water purification and sanitation infrastructure may reduce the diarrheal burden in a marginalized and flood prone population in remote Nicaragua

Authors:

Sheri A Denslow (sheri.denslow@gmail.com)
Jess Edwards (jessedwards@unc.edu)
Jennifer Horney (jenhorne@email.unc.edu)
Rodolfo Peña (rodolfo.pena@epiph.umu.se)
Daniel Wurzelmann (daniel_wurzelmann@med.unc.edu)
Douglas Morgan (douglas_morgan@med.unc.edu)

Version: 6 Date: 7 December 2010

Author's response to reviews:

Edits include formatting changes (spacing) and one word correction.